Author’s response to reviews

Title: Increased Minimum Alveolar Concentration-Awake of Sevoflurane in Women of Breast Surgery with Sleep Disorders

Authors:

yuanyuan cao (caoyuanyuan1220@126.com)
lei zhang (zhlanesth@yeah.net)
xiaohui peng (yirupeng1107@163.com)
yun wu (wuyunanyi@163.com)
quolin zhang (qlzhang@ahmu.edu.cn)
erwei gu (ay_guew_mz@163.com)
Ye Zhang (zhangye_hassan@sina.com)

Version: 2 Date: 19 Dec 2019

Author’s response to reviews:

Dear Professor Tu :

I am so glad to hear from you. Thank you and the reviewers for your suggestions of the revision.

I've made those modifications: 1. The uploaded manuscript and figures are clean version, without highlights or texts in different colours. 2. Additional files which not intended to the publication have been deleted.

Both of two email addresses are my contact informations, but the two addresses are unified recently. So I would like use the "zhangye_hassan@sina.com" when publishing if possible. And sorry for the confusion.

Looking forward to the publication of the article!

Have a nice day, and best wishes!

Ye Zhang